

**Delft University of Technology** 

# Low-head pumped hydro storage

# A review on civil structure designs, legal and environmental aspects to make its realization feasible in seawater

Ansorena Ruiz, R.; de Vilder, L. H.; Prasasti, E. B.; Geisseler, B.; Scanu, S.; Roeber, V.; Marence, M.; Moll, R.; Bricker, J. D.; More Authors

DOI

10.1016/j.rser.2022.112281

Publication date 2022 Document Version

Final published version
Published in

Renewable and Sustainable Energy Reviews

# Citation (APA)

Ansorena Ruíz, R., de Vilder, L. H., Prasasti, E. B., Geisseler, B., Scanu, S., Roeber, V., Marence, M., Moll, R., Bricker, J. D., & More Authors (2022). Low-head pumped hydro storage: A review on civil structure designs, legal and environmental aspects to make its realization feasible in seawater. *Renewable and Sustainable Energy Reviews*, *160*, Article 112281. https://doi.org/10.1016/j.rser.2022.112281

# Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect



# **Renewable and Sustainable Energy Reviews**

journal homepage: www.elsevier.com/locate/rser



# Low-head pumped hydro storage: A review on civil structure designs, legal and environmental aspects to make its realization feasible in seawater

R. Ansorena Ruiz<sup>a</sup>, L.H. de Vilder<sup>b</sup>, E.B. Prasasti<sup>c</sup>, M. Aouad<sup>c</sup>, A. De Luca<sup>d</sup>, B. Geisseler<sup>e</sup>, K. Terheiden<sup>c</sup>, S. Scanu<sup>f,g</sup>, A. Miccoli<sup>d</sup>, V. Roeber<sup>h</sup>, M. Marence<sup>i</sup>, R. Moll<sup>b</sup>, J.D. Bricker<sup>b,j,\*</sup>, N. Goseberg<sup>a</sup>

<sup>a</sup> Leichtweiß-Institut für Wasserbau, Department of Hydromechanics, Coastal and Ocean Engineering, Technische Universität Braunschweig, Braunschweig, Germany

<sup>b</sup> Department of Hydraulic Engineering, Faculty of Civil Engineering and Geosciences, Delft University of Technology, Delft, The Netherlands

<sup>c</sup> Institute for Modelling Hydraulic and Environmental Systems, Universität Stuttgart, Stuttgart, Germany

<sup>d</sup> Department for Innovation in Biological, Agro-Food and Forest Systems, University of Tuscia, Italy

e GEISSELER LAW, Freiburg, Germany

<sup>f</sup> Ocean Predictions and Applications Division, Centro Euro-Mediterraneo sui Cambiamenti Climatici (CMCC), Italy

<sup>8</sup> Laboratory of Experimental Oceanology end Marine Ecology, Department of Ecological and Biological Science (DEB), Università degli Studi della Tuscia, Civitavecchia, Italy

<sup>h</sup> Département of Pyrénées-Atlantiques, L'Université de Pau et des Pays de l'Adour, Pau, France

<sup>i</sup> IHE Delft Institute for Water Education, Delft, The Netherlands

<sup>j</sup> Department of Civil and Environmental Engineering, University of Michigan, Ann Arbor, MI, USA

ARTICLE INFO

Keywords: Pumped hydro storage Renewable energy Low-head hydro Maritime structures

# ABSTRACT

The energy transition requires large-scale storage to provide long-term supply and short-term grid stability. Though pumped hydro storage is widely used for this purpose, regions without natural topography do not have the potential for traditional high-head pumped hydro storage. To address this, multiple projects for low-head and seawater pumped hydro storage have been proposed, though few have been implemented. Here, we review the state of the art of the components of low-head seawater pumped hydro storage projects, for construction in shallow seas or integrated into coastal defenses. We reference all civil infrastructure components, in addition to legal, environmental/biological, and financial constraints, drawing knowledge from proposed, planned, and constructed tidal power and seawater pumped hydro storage projects worldwide. Combining this knowledge, we make a preliminary evaluation of the feasibility for low-head seawater pumped hydro storage in the North Sea. We find that an elevated storage basin is more economical than an excavated one in shallow bathymetry (10 m deep or less), while the reverse is true in deeper water. Corrosion and fouling prevention are already well developed due to implementation of these measures at tidal power plants. Dam construction is feasible if measures are taken to address piping, macro-instability (primarily from rapid drawdown), and bursting of the clay layer. Within the context of Europe, legal and environmental regulations may be the most formidable hurdles to such projects.

# 1. Introduction

Europe aspires to become the first climate-neutral continent in the world by 2050. Already by 2030, the European Union plans to reduce greenhouse emissions at least 55% below 1990 levels [1]. One of the targets defined to achieve this is to decarbonize the energy sector,

i.e. introducing more renewable energy sources into the European grid. Currently, wind, solar, hydropower, solid biofuels, and other renewables are the main sources of renewable energy in the European grid, accounting respectively for 36%, 12%, 33%, 9%, and 9% of all the

E-mail address: J.D.Bricker@tudelft.nl (J.D. Bricker).

https://doi.org/10.1016/j.rser.2022.112281

Received 8 September 2021; Received in revised form 5 February 2022; Accepted 14 February 2022 Available online 4 March 2022

Available online 4 March 2022

1364-0321/© 2022 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

*Abbreviations*: CR, Counter-Rotating; PHS, Pumped Hydro Storage; RPT, Reversible Pump–Turbine; SPHS, Seawater Pumped Hydro Storage; TPP, Tidal Power Plant.; EU, European Union; FRP, Fiber Reinforced Plastic; GRP, Glass Reinforced Polyester; ESOI, Energy Storage On Investment; LCOS, Levelized Cost Of Storage; W, Watt (J/s); H, head (m); Q, volumetric flow rate ( $m^3/s$ ); Rs, length of supercritical flow from breach (m);  $\eta_{gem}$ , average height of flood wave relative to original water level (m);  $a_0$ , average water depth during normal conditions (m); R, distance from breach (m)

<sup>\*</sup> Corresponding author at: Department of Hydraulic Engineering, Faculty of Civil Engineering and Geosciences, Delft University of Technology, Delft, The Netherlands.

renewable energy generated [2]. During the period 2008–2018, wind power, solar power, and solid biofuels technologies have rapidly grown, making wind power the most important renewable source of energy in Europe [2].

The addition of large amounts of wind and solar energy to the grid, due to their stochastic and unpredictable nature, may drive an electric grid system to encounter transmission or operational constraints. At times of large wind and solar energy production, the system operators will have to accept less wind and solar than there is available (curtailment) [3]. The EU's Twenties project [4] showed with market simulations that wind curtailment might increase from 0.4 TWh in 2020 to 9.3 TWh in 2030, due to both increased energy generation and concentration of it at one geographical point.

Lack of transmission capacity is the main factor which governs curtailment, in the case where an excess of wind energy cannot be transported to other areas where it may be used [5]. Grid expansions are expensive and often time-consuming [6]. Energy storage is another option. Instead of taking the energy somewhere else it can be locally stored for later use [7]. Many studies show that wind and solar penetration in the electric grid can be increased with pumped hydro storage (PHS) systems [8–13].

PHS is a mature technology in mountainous regions and comprises 90% of the worlds grid-scale energy storage as of 2020 [14]. Chen et al. [15] showed that PHS technology ranks amongst the cheapest energy storage technologies in terms of costs per kWh of electricity stored and produced. PHS has several advantages, yet large head differences between reservoirs are typically required, rendering countries with lowland topography unsuitable. On a much lower scale of capacity, lithium-ion batteries have made rapid progress toward higher efficiency and lower initial costs, but their lifetime is much shorter and carbon footprint much greater than PHS. For example, Barnhart and Benson [16] showed that the Energy Storage on Investment (ESOI), which represents the ratio of total energy stored over a battery's lifetime to the energy required to fabricate the battery, is 32 for a lithium-ion battery, but over 700 for a PHS facility. Similarly, [17] shows that the Energy Capital Cost distributed over the lifetime of a lithium-ion battery ranges from \$7.5-\$104 per kWhr-cycle, while the cost of PHS is \$0.02-\$1.5 per kWhr-cycle. Therefore, development of PHS feasible for lowland countries would be beneficial for both the environment and the economy.

However, low-head PHS stations (see Fig. 1) are a new technology that have not yet been constructed. The objective of this manuscript is to gather knowledge from previous experience and plans for seawater hydropower (tidal power, high-head seawater PHS and low-head PHS) with the ultimate goal of finding suitable engineering techniques to make low-head PHS feasible. To achieve this, the authors used their expertise to gather information for each section that this manuscript presents. The research started by gathering experiences from operative tidal power stations, experience from the world's only seawater PHS plant (Yanburu SPHS) and plans for other seawater PHS and lowhead seawater PHS plants. This research method is thus limited to the extent of information that the authors could access regarding the described topics. Literature about TPP and high head SPHS is widely available. However, information about low-head SPHS could only be gathered from projects planned in the past and ones being planned at present, as none have yet been built. The remainder of the work is organized as follows: we begin Section 2 with a historical introduction to seawater hydroelectricity. Section 3 focuses on technical aspects of dam construction and dredging processes, Section 4 on technical details for the construction of hydro powerhouses in seawater (special attention is paid to corrosion and fouling) and Section 5 deals with the state of the art of low-head reversible pump-turbine technology. Section 6 deals with environmental and legal aspects to be considered for the development of low-head PHS. In Section 7 an analysis of costs based on the reviewed literature is done. Section 8 discusses several topics that could lead to future research on low-head PHS. Finally, Section 9 concludes with the main findings from this review.

# 2. Past projects combining seawater and pumped hydro storage

We will begin our review by critically analyzing the interconnected lessons of PHS involving seawater. Seawater pumped hydro storage (SPHS) grows out of two existing technologies: high-head PHS and seawater tidal energy generation. High-head PHS encompasses 160 GW of installed capacity worldwide as of 2020 [14]. Its economical and ecological advantages warrant further research into PHS as a promising technology in appropriate settings. The International Hydropower Association estimates the installed capacity will increase to about 240 GW by 2030 [27].

# 2.1. Tidal power

SPHS for lowland countries can employ early knowledge from lowhead tidal hydropower projects, since these have been operating hydroelectric equipment in seawater for several decades.

The La Rance tidal power plant (TPP) was the first modern TPP when its operation began in 1966, having an installed capacity of 240 MW. It is a 750 m long barrage that blocks the Rance River estuary at its narrowest point in St. Malo, forming a 22 km<sup>2</sup> basin [28]. The 1.7 MW TPP at Kislava Guba (Kislogubsk), in experimental operation since 1968, was the first TPP to use floating caissons for the powerhouse construction in a narrow strait of 50 m width close to Ura Bay [29]. The Jiangxia TPP, currently the largest experimental TPP, started operating in 1980 and was integrated into a 670 m long existing rockfill dam [22]. The Annapolis TPP, the first TPP in Canada, was built in 1984. Its purpose was to demonstrate the commercial operation of a 7.6 m diameter straight-flow (Straflo) turbine. The choice of the turbine diameter for the Annapolis TPP was aligned with the turbine diameters used in the large-scale feasibility studies in Fundy Bay [30]. In 2011, the Sihwa TPP surpassed La Rance as the largest TPP with an installed capacity of 254 MW. A key aspect for the site identification of the Sihwa TPP was the possibility to flush the polluted Sihwa reservoir [31]. Characteristics of the above-mentioned TPP's are shown in Table 1. Meanwhile, many tidal sites worldwide have been investigated as possible TPP locations e.g. Severn (U.K.), Swansea Bay (U.K.), San Jose (Argentina) and Bay of Fundy (Canada). Efforts to improve existing TPPs are still ongoing. These early developments were usually based on topographic features, sometimes enhanced by civil works, to form specific TPP reservoirs for storage of potential energy supplied through tidal motions.

# 2.2. High-head seawater pumped hydro energy storage

By 1999, the Okinawa Yanbaru SPHS Power Station started operation and became the first PHS facility in the world to use seawater to store energy [32]. The installed capacity and storage capacity is shown in Table 2 together with the rest of the SPHS stations discussed in this section. The plant's cost was 30 billion Yen (around  $\in$ 232 million today) [33]. During the first five years of operation, the station was used as a research facility. Later, it provided electricity to Okinawa island until it was dismantled in 2016 because the energy demand on the island did not grow as expected at the beginning of the project [33]. Similarly to La Rance TPP, the acquired experience of Okinawa SPHS was a milestone for later plans.

Currently, several other projects have been planned but not constructed yet [10-12,34-37]. In Ireland, the Glinsk seawater pumped hydro was planned in 2012, involving a 960 MW station that could store up to 6 GWh per year. Its construction was planned to happen between 2014–2017 [38], yet not realized to date.

Several other plans have been developed for energy independence based on renewables in the Greek islands [10,34,35]. Katsaprakakis et al. [34] proposed two seawater pump hydro storage projects on the islands of Creta and Kasos. Partnering with the electrical contractor ENET S.A, they showed that installing wind parks together with SPHS



Fig. 1. Schematization of a low-head PHS scheme. Powerhouse conceptual design from: Ansorena 2020 [18]. Notice that the selected turbines are of the Francis type.

Overview of relevant tidal power plants and their key features.

	La Rance	Kislaya Guba	Jiangxia	Annapolis	Sihwa
Location	France	Russia	China	Canada	South Korea
Area of reservoir (km <sup>2</sup> )	22 [19]	1.1 [20]	1.5 [ <mark>21</mark> ]	6 [22]	42.44 [23]
Volume of reservoir (Mio. Mm3)	184 [ <mark>19</mark> ]	-	5.14 [ <mark>21</mark> ]	-	147 [ <mark>23</mark> ]
Date of commissioning	1966	1968	1980	1984	2011
Operator	EDF	RusHydro	-	Nova Scotia Power Inc.	Kwater
Installed generator power (MW)	240 [19]	0.4 [20] after 2006: 1.7 [24]	4.1 [ <b>21</b> ]	20 [25]	254 [23]
Yearly energy output (GWh)	540 [19]	1 [20] after 2006: -	7.2 [21]	50 [25]	552 [ <mark>23</mark> ]
Mean tidal range (m)	8 [19]	2.4 [26]	5.08 [21]	6.5 [25]	5.6 [23]

Table 2

Characteristics of high-head seawater PHS projects reviewed. Notice that the upper reservoir of Kasos has a much larger storage capacity than necessary for daily storage. This is because of the topography of the island. The authors argue that this is an advantage for: (a) operation of the wind powered PHS for periods other than at peak power demand, (b) contribution of the wind powered PHS to controlling power production, and (c) improvement of the system's security.

PHS	Okinawa	Glinsk	Cultana	Rhodes	Crete	Kasos	Espejo de Tarapacá
Year of planning	1999	2012	2013	2014	2013	2013	2021
Installed power (MW)	30	960	225	160	64.5	4	300
Head difference, average (m)	142	292	260	270	500	472.5	604.75
Storage capacity (MWh)	188	6100	1770	1603	2346	516	35400

is a promising technology for insular systems with low annual rainfall [34]. On the island of Rhodes, a 150 MW plant could reduce the annual wind curtailment by 75% [10]. The most promising Greek case is the SPHS proposed for Sifnos island, which is planned to help achieve 100% local renewable energy generation together with an offshore wind park [35]. The project is currently looking for financing. In Australia the Cultana PHS project resulted from collaboration between the firm ARUP and the Australian Renewable Energy Agency. However, due to financial reasons, the plan did not go forward [39]. ARUP wanted to purchase the public land for installation of the plant but the Australian Government withheld permission [40]. Finally, in Chile the Espejo de Tarapacá project considers a 300 MW plant. Initially, the plan was to start construction in 2020. Currently, the plan is looking at finance methods and they aspire to have an operating plant by 2026 [41].

# 2.3. Low-head seawater pumped hydro energy storage

Low-head PHS is the most innovative SPHS technology and is necessary if PHS is to be feasible in countries without natural elevation differences available. Few plans have been developed so far, yet none have been constructed. In 1981, the Dutch engineer Luc Lievense published a proposal for an energy storage lake in the Markemeer (a large lake in the Netherlands) [42]. His plan involved the installation of wind turbines to pump water inside a 14 m high (relative to sea level) dyke ring. The water inside the ring would rise to 12 m above sea level and it would be turbined out of the basin when electricity is needed [43]. Afterwards, the plan developed into various alternatives at different Dutch locations such as the Haringvliet, Brouwersdam and the IJsselmeer. The final plan ended up with a 70 m high dyke alternative at the Brouwersdam [44]. Following the Lievense plan, in 2007 KEMA consulting and Lievense BV developed a new energy island design [45]. This design included a dredged 40 m deep basin [45] to be used as a low reservoir, with the sea as the upper basin. This 'bathtub' concept was named a 'valmeer', with the benefit of lower dykes avoiding safety concerns related to dyke breaching, which was a worry that plagued the 1981 plan.

In Taiwan, another design named TIESI (Taiwan Integrated Energy Storage Island) was developed in 2014. There, the water level inside the basin remains always below sea level just like the KEMA consulting plan. This plant considered the use of Francis turbines, which are a preferred option for all low-head PHS systems reviewed in this paper. However, Francis turbines are not ideal for low-head applications below 20 m (See Fig. 2), while axial Kaplan and propeller devices have not yet been optimized as reversible pump–turbines [46]. For instance, La Rance's axial Kaplan (bulb) reversible pump–turbines, designed by Neyrpic France in the 1960's, have a hydraulic efficiency for turbining and pumping of 85% and 65% respectively [47], leaving room for improvement.

In 2018, the Belgian Federal Government financed the planning of a multifunctional island (iLand) which includes the following functions: energy storage (low-head PHS), renewable energy generation



Fig. 2. Application range of Voith Hydro turbines. *Source:* Voith GmbH webpage [56].

(installing new wind turbines and floating PV panels) and marine aquaculture. The plan's execution has been delayed due to permitting issues [48]. Since this would be the first cross-border hybrid storage project of its kind, negotiation with many different authorities is delaying the execution of this plan more than expected. Additionally, the regulatory framework concerning storage currently carries a rather large degree of uncertainty. It is expected that uncertainty will be reduced with the forthcoming adoption of the so-called "Clean Energy for all Europeans package" [48]. Some low-head PHS such as Delta21 [49] are integrated in the flood protection system of a country. Delta21 is an integrated plan for flood safety and energy storage in the area of the Haringvliet (South Holland, The Netherlands). Flood safety is provided by the construction of a new storm surge barrier and the use of a pumping station. The pumping station will be used to evacuate the river discharge during severe storms when the storm surge barriers are closed and river flow accumulates upstream of the barrier. To take advantage of the pumping station, a storage lake is constructed around it to have an energy storage basin which can be used daily. The Delta21 design considers 1920 MW of power with a storage capacity of 14.4 GWh. This project is currently studied on its feasibility. [18,50-55]. Table 3 shows the main characteristics of the analyzed low-head SPHS projects.

# 3. Dam design

This chapter elaborates on the characteristics of dam structures by first analyzing experience from tidal power construction to then continue with SPHS. For SPHS the main choice between a high or low reservoir is explained and the most critical failure mechanisms identified.

# 3.1. Tidal power

High-head PHS often uses Francis turbines for energy generation, but these are not designed for low-head operations (Fig. 2). In contrast, tidal energy stations obtain energy from low-head operation, thus being a reference for low-head seawater PHS.

# 3.1.1. Site characteristic and design

Potential TPP sites are examined by taking into consideration the average tidal range, the feasibility of the plant construction and the environmental impacts [22]. In addition, the design of a TPP depends on the purpose of the construction and potential secondary usage (e.g. road works). Therefore, the constructed TPPs have different scales. La Rance TPP operates on a commercial scale providing cost efficient energy for the bordering regions of the plant (1.8  $\in$  cents/kWh in

2009 [57]). Consequently, a large tidal range is essential for the La Rance site. The purpose for constructing Kislaya Guba TPP was to investigate the applicability of the floating powerhouse construction method on a large scale. Its location, even though it does not have a very large tidal range, was chosen due to the shape of the basin and the proximity to the grid [29]. The Annapolis TPP location at the mouth of the Annapolis river was selected mainly due to the existing causeway and sluice structure built in 1960 [30]. In addition, the existing tidal range in Annapolis was suitable for testing the application of a large diameter Straflo turbine [58]. Among the numerous small-scale TPPs in China, the Jiangxia TPP is the largest in operation. Most of the TPPs in China were decommissioned due to an inappropriate location and/or outdated technology [59]. However, the approach of small-scale TPP constructions in China was realistic and effective [60], contrary to the planned large-scale projects. In South Korea in 1994, a 12.6 km barrier was constructed to create the fresh water reservoir Lake Sihwa for providing irrigation water and additional land. Since the wastewater of nearby industry severely polluted the reservoir, the government later decided to enable water circulation between the reservoir and the sea and to simultaneously generate electricity [31].

# 3.1.2. Embankment construction

A TPP is generally composed of a powerhouse, a sluiceway, embankments and a navigation lock. A road is usually located on top of the dam to connect both ends of the estuary, providing added benefit to the overall plant. To date, all constructed TPPs follow the tidal barrage technique [61], which consists of closing the estuary at a narrow location with a powerhouse and an embankment. TPP designs are constrained by investment costs, which are strongly related to the chosen construction method [62]. At La Rance the main cost factor was the cofferdam construction [20]. The use of a cofferdam (dry construction method) was chosen for the dyke and powerhouse construction, to control high discharges during tides due to the closing of the bay [28]. The Kislaya Guba experience showed that the floating powerhouse construction method (wet construction method) can cut the cofferdam cost [20]. The floating method reduced the costs compared to conventional methods with cofferdams by one-third [63]. However, the floating method requires the preparation (dredging) of a channel, which can become expensive in shallow waters. In the Annapolis TPP, the rockfill causeway was already in place. The Annapolis TPP powerhouse was constructed on Hog's Island using the dry method. For Sihwa TPP, the integration of the powerhouse in the embankment dam was done using the dry construction method with a cofferdam and enclosure.

Sealing the embankment dam of the TPP is advantageous since it secures the construction and prevents head losses due to seepage.

Characteristics	of reviewed	low-head	PHS	projects.	

PHS	Lievense Plan	KEMA	TIESI	iLand	Delta21		
Country	The Netherlands	The Netherlands	Taiwan	Belgium	The Netherlands		
Year of planning	1986	2007	2014	2018	2021		
Installed power (MW)	1500	1500	1500	550	1920		
Storage Capacity (GWh)	30	20	20	2.2 (10 fn)	14.4		

The 163 m long rockfill dam in La Rance has a concrete core for sealing, with an inspection gallery to measure deformations [64]. A rock revetment was placed for protection against wave and tidal actions. The Annapolis studies showed that allowing leakage through the existing causeway could significantly increase the costs of the project. Therefore, sealing of the causeway was carried out using the glacial till removed from the excavations on Hog's Island, allowing simultaneously the disposal of the till [58]. The dam in Jiangxia has a clay core for sealing purposes [21].

# 3.2. Low head pumped hydro storage

After a brief introduction to the key characteristics of offshore PHS this paragraph discusses the choice for creating an upper or lower inner reservoir supported by an economic analysis to determine what configurations of low-head seawater PHS are appropriate for various water depths. Furthermore the potential flood risk is discussed as well as construction materials, measures against seepage and the most important failure mechanisms.

# 3.2.1. Main characteristics of offshore PHS

A constructed (circular or otherwise enclosed) dam enables energy storage by controlling the water level of the inner reservoir and retaining a head difference between the out- and the inside. The energy storage capacity (E) of a PHS reservoir is determined by the head difference and the surface area (Eq. (1)). For a valmeer, the dam costs scale linearly with the circumference.

$$E = mgh = A\rho g \frac{H_{max}^2 - H_{min}^2}{2} \tag{1}$$

Where:

- E = Storage capacity (J)
- m = Mass of water (kg)
- g = Gravitational acceleration (m/s<sup>2</sup>)
- $\tilde{h}$  = Height difference (m)
- A = Area of the basin (m<sup>2</sup>)
- $\rho$  = Density of water (kg/m<sup>3</sup>)
- $H_{max}$  = Maximum head difference between upper and lower basin(m)
- *H*<sub>min</sub> = Minimum head difference between upper and lower basin (m)

Therefore, the storage capacity scales with the diameter squared, whereas the circumference scales linearly with the diameter. Hence, the storage costs,  $\in$ /GWh, decrease for a larger reservoir. Furthermore, the storage capacity scales with the head difference squared. To store a certain amount of energy a larger head difference results in a smaller reservoir diameter, which decreases the overall dam costs (not the dam costs per meter). This effect can be seen in Fig. 3. For the dam costs, the dimensions from Fig. 4 are used. The dredging costs are estimated at  $\in 8 / m^3$  and the dredged volume is expected to be 1.4 times as large as the dam volume to account for spillage and settlement. Additionally,  $\in 60 / m^2$  is considered for the dam footprint for drainage and soil preparation and  $\in 60.000/m$  for revetment costs per meter of dam length.



**Fig. 3.** Estimated total dam costs for seawater PHS, situated in 20 m deep water and equipped with bank protection to resist up to 8 m high waves (De Vilder, 2017). The calculations are done with the assumption that the Counter Rotating (CR) propeller reversible pump-turbine (RPT) operates down to an  $H_{min}$  of 2 m, while the Francis RPT operates from the shown maximum head difference down to a minimum head difference of  $H_{max}/1.25$ .

## 3.2.2. Upper or lower reservoir

The choice to use the sea as an upper versus a lower reservoir is straightforward for schemes that use the natural height difference along a coastline with cliffs. However, for offshore facilities both options can be considered. Studies by Lievense ended up with a preferred alternative that contained an inner reservoir that was 70 m higher than the sea level [44]. In 2007 KEMA and Lievense came up with the inverted process that would use the inner lake as the lower reservoir, called a 'valmeer' [45]. Since then all offshore studies have been based on the 'valmeer' principle, because it significantly reduces the dam volume [49,50,65].

3.2.2.1. Volume balance. As an example of the difference in dam volume, see Fig. 4. Both dams are able to store the same amount of energy, but the valmeer alternative only requires about  $\frac{1}{3}$  of the dam volume. However, Fig. 4 does not show the amount of material that needs to be dredged for the valmeer. In case a 40 m head difference is created in 20 m deep water, roughly 400 000 000 m<sup>3</sup> of material is leftover. In order to realize a feasible project it is therefore paramount to take the volume balance into account, to not end up with either a big deficit of construction material or surplus of dredge spoils. The relation between water depth, redundant material, storage capacity and maximum head difference is shown in Fig. 5.

The volume balance from Fig. 5 shows that for a certain water depth it is not feasible to construct a reservoir with any kind of storage capacity and head difference. From line 2 (orange) in Fig. 5 it can be deduced that for 10 m deep water a reservoir with 20 GWh of storage capacity and 20 m of maximum head difference results in almost  $1 \times 10^9$  m<sup>3</sup> of excess material. In case the valmeer could be dredged for  $\in 8/m^3$ , this would lead to  $\in 8$  Bn. of additional costs. In case a valmeer would be constructed offshore in combination with large-scale wind power development, it could be integrated with the creation of an 'energy island' for all the electric infrastructure and the operation and maintenance of the wind farms. Such an island could require  $2 \times 10^8$  m<sup>3</sup>



Fig. 4. The maximum water level difference ( $H_{max}$ ) of 40 m is created for the valmeer (left) by dredging 20 m from the inner reservoir. The dam with the high inner reservoir (right) creates a 40 m head difference. The minimum head difference ( $H_{min}$ ) is indicative and can change per turbine type.



Fig. 5. The volume balance of both a valmeer and a high reservoir PHS facility for a given storage capacity (E) and head difference ( $H_{max}$  and  $H_{min}$ ). Lines 1 to 6 use a Francis RPT and lines 7 to 9 use a CR RPT. The vertical axis shows the amount of redundant material, from which the dam volume is already extracted. Note that the dam is assumed to require 1.4 times its volume for dredging to account for spillage and settlement.

of sand [66], which can be sourced from the storage lake. Therefore in general a 'valmeer' type reservoir is more attractive for larger water depths (20 to 30 m) and a high reservoir in shallower water (5 to 10 m). Fig. 5 also shows that application of the CR RPT results in a broader water depth range in which the volume balance remains manageable than application of Francis turbines. This is caused because CR RPTs operate over a larger water level range at high efficiency, which reduces the reservoir surface area required to store the same amount of energy.

*3.2.2.2. Flood risk.* Aside from volumetric considerations, flood potential is another important aspect, as breaching of dam sections could pose a significant risk. Where a dam failure of a valmeer mainly incurs economic loss, a breach of a high reservoir could cause an extreme flood event, thereby requiring a higher design safety level than required for a valmeer. This effect is illustrated by the design failure probabilities that were calculated for high storage reservoirs near the Brouwersdam (integrated in coastline) and in the IJsselmeer (interior lake) [44].

For these 70 m high reservoirs, models show that the maximum outflow discharge could become 277 000  $\text{m}^3/\text{s}$  (130 000  $\text{m}^3/\text{s}$  and 99 000  $\text{m}^3/\text{s}$  for 40 m and 24 m head difference respectively) in case of an instantaneous dam breach over the whole height of the dam and with a progressing breach width. From the breach a supercritical flow would develop that after a hydraulic jump transitions into a flood wave. The length of the supercritical flow and the average height of the flood wave are computed as follows:

$$R_{s} = \sqrt{\frac{\left(\frac{Q_{max}}{\pi}\right)^{2}}{gf^{3}}}$$
(2)

$$\eta_{gem} = 0.55 \frac{Q_{max}}{\sqrt{ga_0}R} \tag{3}$$

Where:

- $R_s$  = Length of supercritical flow from breach (m)
- $Q_{max}$  = Maximum discharge through breach (m<sup>3</sup>/s)
- g = Gravitational acceleration (m/s<sup>2</sup>)
- $\tilde{f}$  = Coefficient for bottom friction: 0.004 (-)
- $\eta_{gem}$  = Average height of flood wave relative to original water level (m)
- $a_0 = \text{Average water depth during normal conditions (m)}$
- R = Distance from breach (m)

This Brouwersdam scenario involved the failure of a protective dyke or dune subjected to the 1.7 km long supercritical flow [67]. The subsequent 7 m to 11 m high flood wave would evenly inundate the surrounding polders. Based on the inundation depth and the site's population, the expected death toll of a dam breach was estimated as 313 persons. Most casualties (244 persons) would occur due to the high speed of the supercritical flow, which was deemed 5 times as deadly as the subcritical portion of the flood wave. The total expected casualty rate led to a required design failure annual exceedance level of  $10^{-4}$  or less. The same analysis for the IJsselmeer location resulted in a higher expected death toll of 3735 and a corresponding maximum allowable failure probability of  $10^{-6}$  pear year, due to the deeper surrounding polder and denser population.

It was found that failure due to erosion of the dam crest due to overflow was the largest threat for the whole system [67]. This would be caused by a lack of control of the maximum upper water level. Consequently, it was suggested to use 2 or 3 independent monitoring systems to make the failure probability sufficiently small. Additionally, it was thought that for the coastal application of a high reservoir it could be socially beneficial to make the dam higher on the land-side so in case the dam would fail, it would likely flood towards the sea-side.



Fig. 6. On top, the dune design with a shallow foreshore from the valmeer project Delta21 that would be integrated into the coastline (modified from [72]). Below, the dam design for the offshore valmeer project iLand that uses hard armor protection and includes a cement–bentonite wall to limit seepage through the dam (modified from [65]).

The design failure probability of the valmeer that is considered for the Delta21 project would need to be at least  $10^{-3}$  to be incorporated into the coastal flood protection network, to equal the failure probability of the current Haringvliet sluices that it would replace [68]. In general, an abrupt dam breach of a valmeer results in a negative flood wave [69,70], which due to the sudden drop in water pressure, could potentially cause outer slope stability failure of surrounding dykes. The added risk of inundation of a polder, caused by a dam breach that induces outer slope failure can be neglected. Consequently, it seems reasonable to assume that the risk of flooding caused by a valmeer's dam breach does not exceed the personal acceptable risk level in the Netherlands of  $10^{-4}$ .

# 3.3. Dam structure

There is a large variation in the types of dam structures for low-head PHS projects, see Fig. 6. When there is no natural height difference available, some schemes use the existing coastline to reduce costs or offer additional services like flood protection [49]. All considered offshore dams use earth based designs, consisting of dredged material from the inner reservoir. For high reservoirs alternative designs were researched, using crown wall structures made of reinforced earth of roughly  $\frac{1}{3}$  of the reservoir's height [71]. The use of a crown wall would significantly reduce the dam volume and could therefore save on the construction costs.

A low inner reservoir requires construction on top of a thick impermeable (clay) layer that prevents upburst and limits seepage through the bottom of the reservoir [45]. For the high inner reservoirs, considered in the Plan Lievense, seepage through the bottom was sufficiently reduced by depositing a 4 m thick and 200 m wide layer of marine silt at the inner toe of the dam [71]. Measures against seepage through the dam for low reservoirs consist out of bentonite walls [45]), sheet pile walls or impermeable cores [18,50,54] and for high reservoirs out of membranes and asphalt lining on the inner slope [36,37,71].

Protection against incident ocean waves is either provided by a dune profile [44,49] or a hard armor protection with appropriate freeboard as necessary to restrict wave overtopping [50,65]. The choice for hard or soft bank protection is not only based on economic aspects, but also on landscape and ecological integration as well as recreational aspects [71]. Therefore schemes that would be integrated in the coastline will likely present a dune profile while offshore alternatives will more likely contain rock or concrete armoring.

Failure mechanisms of a dam including instability, piping and liquefaction are considered manageable when adequately designed for [50, 67]. Van Adrichem [52] found that SPHS dam stability can be greatly affected by the rapid drawdown of the inner water level when the dam is saturated. He found that the slope of the Delta21 sand dune could be reduced to 1:5 and still be stable when considering several cycles of drawdown (17.5 meters of head drop/gain in 12 h). From the literature, it was found that stability during the construction process, dam settlement, and the constructability of dams in offshore conditions have not yet been investigated in detail. For the dam design of a valmeer special attention needs to be paid to the inner slope stability considering the uplift mechanism at the toe [50]. A distinctive mechanism that needs to be taken into account for a high reservoir near the coastline is salt intrusion, caused by 'far seepage' from the high water pressure inside the reservoir [44]. Overall no technical showstopper has been encountered for the dam construction.

# 4. Powerhouse and conveyance system

### 4.1. Powerhouse structure in tidal stations

Worldwide, TPP powerhouses have a similar structure, formed as a hollow concrete structure consisting of an upper and lower part. This structure can be referenced for low-head SPHS construction. The upper part mainly includes a maintenance hall with hoisting equipment. The lower part is laterally divided into subparts, each containing one turbine unit. Each turbine unit is connected to an inlet and a tailrace forming a straight waterway, which enables water passage in both directions with high efficiency. The flow is controlled by a wicket gate, a component of the turbine. To allow maintenance on the turbines and waterways under dry conditions, the powerhouse is equipped with a drainage system as well as gates or stop logs at the end of each inlet and tailrace. Since the waterways are not connected to each other, maintenance work can be done individually for each turbine and waterway.

Despite the similarities between the reviewed TPP powerhouses significant differences emerge from the powerhouse dimensions (Table 4), the waterway shapes, the placement of the turbine units within each powerhouse, and the concrete composition of the powerhouses. The differences among powerhouses are mainly due to the TPP capacity, the mode of operation, the highest tide level, the construction method, and the site conditions.

TPP Powerhouse characteristics.

	La Rance	Kislaya Guba	Annapolis	Sihwa
Powerhouse Dimensions <sup>a</sup> :	53.4 [73]	36 [74]	46.5 [30]	61.1 [23]
- Width <sup>b</sup> (m)				
- Height <sup>c</sup> (m)	25 [73]	3.25 [74]	23 [30]	29 [23]
- Length (m)	390 [73]	18.3 [74]	25 [30]	380 [75]
Operating mode	Two-way generation	Two-way generation	Single-action, ebb generation	Single-action, flood generation
Construction Method	Dry	Wet	Dry	Dry
Number of Turbine(s)	24 units	2 units	1 unit	10 units

<sup>a</sup>Approximation.

<sup>b</sup>The start of the penstock to the end of the tailrace.

<sup>c</sup>Above the foundation level.

# 4.2. Construction methods for tidal power stations

TPP powerhouses are built either using the dry construction or wet construction method, depending on the site characteristics [76]. Dry construction methods require a cofferdam and consequent dewatering of the construction site, while the wet construction does not demand such requirements.

For the dry construction method cofferdams are constructed with different techniques. The cofferdams of La Rance TPP were built on the upstream and the downstream side using precast concrete caissons and sheet piles [73]. However, the cofferdam for the Sihwa TPP was built using long cylindrical cells, which are connected by spandrel walls [23]. The cofferdam for the Sihwa TPP was constructed only on the sea side, since the existing dam in Sihwa served as a cofferdam on the basin side. The dry construction of Annapolis TPP was less sophisticated, making use of Hog's Island at the mouth of the Annapolis River as a construction site. The powerhouse was constructed at Hog's Island by excavating a construction pit. The glacial till from the excavation was used to build the cofferdam on the upstream side of the island [25].

The construction of a TPP powerhouse in the dry is typically conducted according to the following steps: foundation preparation, powerhouse erection, and mechanical and electrical components installation. TPP powerhouses are generally made of concrete with high resistance against seawater [64,73,77]. In La Rance TPP, blast furnace cement (with 75% furnace slag) was used to increase the durability of the powerhouse [64]. The heavy components such as turbine units are installed using hoisting equipment. After the completion of the TPP powerhouse, the cofferdams are dismantled.

For the wet construction or floating method, the powerhouse is pre-fabricated and subsequently transported by water to the construction site. This method was used for the powerhouse of Kislaya Guba TPP [77]. In order to obtain reliable contact with the foundation, the bottom of the Kislaya Guba powerhouse was fitted with shear keys which were buried 25 cm deep in the sand layer [29].

# 4.3. Powerhouse structure in seawater pumped hydro and construction methods

High-head pump stations require the pump to be positioned deep below the lower reservoir water level in order to avoid cavitation. In Table 5 installation depths are shown. The high-head SPHS plant in Okinawa has an underground powerhouse, which can be accessed through a vertical shaft. This typical set-up was also proposed for Glinsk [37] and Espejo de Tarapacá [36]. For those projects, the powerhouse hosts reversible pump-turbines. In contrast, projects such as the Greek SPHS proposals consider the construction of separate stations for pumping and turbining [10,34,35]. According to the authors [10] this allows flexible operation which maximizes wind energy penetration and contributes to the system's stability and dynamic security. The total annual averaged efficiency of this system is 68.95% [34]. Both stations would be constructed at sufficient distance from the coastline to avoid direct contact with seawater [34] and excavation costs. The pumping units of the Greek SPHS schemes are considerably smaller than the rest and thus their required installation depth is also smaller (see Table 5).

In the Japanese SPHS project, a concrete tetrapod breakwater was constructed to protect the water intake. High wave energy in the proximity and minimal distortion to local water currents supported the construction of this structure. The Glinsk SPHS considered a similar design in which a tetrapod/Xbloc breakwater was considered both for protecting the inlet from the waves and currents and to dissipate the energy of the water entering the sea during the turbining operation [37]. For the Greek SPHS, the authors recommended building a submerged breakwater to avoid visual effects. The Espejo de Tarapacá seawater inlet has a diameter of 5 m which is covered by a 16 m diameter trash rack built of 1 cm bars with a spacing of 5 cm which aims to prevent large debris from entering the inlet pipe. Table 5 lists seawater intake depths ranging between 15–20 m for most projects proposed after Okinawa's plant due to the weak currents and insignificant wave motion at this depth.

The Lievense plan considered an in-situ built PHS station. The building pit was to be dredged to a depth of -25 m NAP (Normaal Amsterdams Peil). The dredged material is used to build the dyke of the building pit. Then, the building pit is drained and excavation works in the dry are carried out until reaching the level -40 m NAP. Finally, the powerhouse is built in the dry environment provided by the building pit. TIESI also considered the dry construction within a building pit for the PHS powerhouse construction. On the contrary, the iLand project considered modular construction with the use of prefabricated elements [45].

The Delta21 project studied several alternatives from in-situ built massive concrete structures to a modular prefabricated caisson structure positioned with the help of tugboats, considering the construction of a smaller PHS in combination with a sand dune [18,78]. Paasman and Ansorena [18,78] found that piping protection is important for the building pit construction, due to the difference in water head with both sides of the building pit's dyke. Piping solutions such as sheet-pile screens and the use of impermeable geotextiles, among others, are available.

An impermeable soil layer (e.g. clay) can affect the maximum excavation depth of a building pit or inner basin. When the clay layer is close to the surface, a sheet pile wall can be driven under the dam all the way down to the impermeable layer. This will restrict seepage into the reservoir and piping under the dam. However, the closer the impermeable layer to the surface, the shallower the maximum excavation depth becomes. This is due to the water pressures acting under the impermeable layer (See Fig. 7). Note that this upburst mechanism is also the limiting factor for the head difference inside a valmeer [18]. For a maximum excavation depth, the larger the pump-turbines, the lower the maximum water level difference available in the valmeer. Table 6 shows different submergence values for the pump-turbines considered by Ansorena (2020) [18] for the Delta21 powerhouse. Notice that the larger each pump-turbine unit, fewer units are needed to provide the desired power for a low-head SPHS plant, thus reducing the overall powerhouse width (material reduction).

## R. Ansorena Ruiz et al.

#### Table 5

Characteristics of the intake pipe and submergence depth of the reversible pump-turbines in the high-head SPHS schemes analyzed.

11 0	1	1	1	0		<i>,</i>		
	Kasos <sup>a</sup>	Rhodes <sup>a</sup>	Crete <sup>a</sup>	Curaçao	Okinawa	Espejo de Tarapacá	Cultana	Glinsk
Single pump/turbine power (MW)	0.560	1.074	1.162	19.8	30	100	117	300
Head difference (m)	472.5	270	500	-	142	604.75	260	292
Installation depth below sea level (m)	2	1	1	18-21	25	40-45	-	40-45
Intake pipe (seawater) length from coast (m)	635	20	400	-	27	340	-	-
Intake pipe (seawater) installation depth (m)	15	18	20	-	8	15.5	-	15-20

<sup>a</sup>Notice that the systems of Rhodes, Crete and Kasos consider separate structures for pumping and turbining. Therefore, for those cases the units are pumps instead of pump-turbines.

# Table 6

Powerhouse dimensions and submergence for different pump-turbine characteristics. Submergence is defined as the depth from the lowest water level of the valmeer to the pump-turbine position.

<i>Source</i> . Ansorena (2020) [16].							
-	Alternative 1	Alternative 2	Alternative 3	Alternative 4	Alternative 5		
Pump discharge (m <sup>3</sup> /s)	27	60	100	160	200		
Pump power (MW)	5.0	11.1	18.5	29.6	37.0		
Submergence (m)	2.4	3.6	4.7	5.9	6.6		
Width estimation of Powerhouse <sup>a</sup> (m)	3320	2228	1722	1372	1218		

<sup>a</sup>Based on the number of necessary pump-turbine units times the width of a single unit.

Note that these values are obtained for a head difference of 14 m, the average head difference of the Delta21 valmeer, which uses Francis pumps from Pentair/Nijhuis as reversible pump-turbines.



**Fig. 7.** Representation of soil upburst failure mechanism. When the water pressure under the impermeable soil layer is not countered by the weight of material and water above it, the water pressure will burst upward through the clay, causing failure of the building pit and its consequent flooding.

## 4.4. Corrosion prevention measures

A common corrosion prevention method used in most TPP powerhouses is active cathodic protection. In the La Rance TPP, in addition to stainless steel construction and painting of the turbine runners, active cathodic protection is used for the 24 turbine units, 6 sluice gates, and the metallic parts of the shipping lock. The electric power required for the supply of the active cathodic protection at La Rance TPP is 20 kW, with an annual consumption of 150,000 kWh [79]. In the Kislaya Guba TPP, the total power required for the active cathodic protection of the powerhouse is 3 kW, with annual consumption less than 12,000 kWh. Apart from active cathodic protection, stainless steel was also used in the Annapolis TPP for turbine blades and rotor rims [30]. Investigations after one year of operation (with 99% availability rate) showed minor spots of pitting corrosion on the runner hub. This demonstrates the effectiveness of active cathodic protection [25].

Corrosion prevention in Okinawa was different for components subjected to rapid vs. slow flow speeds [80]. Mild carbon steel coated with paint was used for the lower flow velocity regions whereas stainless steel was used for high flow velocity portions. Active cathodic protection was also used to prevent corrosion due to paint damage and against crevice corrosion. Since corrosion accelerates with higher flow velocities, the cathodic protection was designed with an adjustable protective electrical current. The wicket gate and runner were made of austenitic stainless steel with a low carbon content plus nitrogen, which was added to improve corrosion resistance. The main shaft – made of forged stainless steel – was equipped with a slip ring to provide the current cathodic protection. The draft tube was made of two different materials. The upper part was made out of austenitic stainless steel with a low carbon content whereas the remaining part was made of rolled steel which then was coated with a thick film of paint (vinyl-ester-like resin) containing glass flakes. Finally, the main shaft's sealing box was ceramic. In addition, the penstocks were made of Fiber Reinforced Plastic (FRP) with rubber joint seals, to avoid corrosion and to deal with the high flow speeds and pressures.

For relatively small diameters, [34] recommends the use of Glass Reinforced Polyester (GRP). The chemical structure of this material is inert to seawater. Moreover, its surface is smooth to reduce the head loss in the pipe system. It is both lighter and cheaper than steel, which also makes its installation simpler and thus cheaper. The drawback of this material is that the larger the pipe diameter, the lower the allowable internal water pressure. In the case of Kasos, a 0.8 m diameter pipe is considered. This limits the pressure to 32 bar [34]. In Crete, the diameter is 2 m and the maximum pressure drops to 10 bar [34]. GRP pipes can be combined with steel pipes if higher pressures are required. The interior of the steel pipe used in Kasos is covered with a thick film of mixed phenol and epoxy resins, without solvent [34].

Measures to avoid corrosion of the materials can be very expensive. The Australian seawater PHS system found that it was more profitable to build a desalination plant and use fresh water in the system than running it with seawater due to future maintenance costs [40]. The costs of constructing the water desalination plant plus the lower reservoir were lower than the costs of applying corrosion preventive measures over the lifetime of the structure.

# 4.5. Antifouling measures

Marine fouling, defined as the unwanted accumulation of marine organisms immersed in the sea [81], is an important factor both from economic [81,82] and environmental perspectives.

Fouling related problems generally occur in TPP powerhouses, causing deterioration of the structure and enabling settlement of organisms e.g. algae, mussels, which can partly clog the waterway. To prevent fouling, antifouling coatings and regular cleaning are used in all TPPs. [83,84] showed that fouled surfaces require recurring maintenance and cleaning activities, which constitute a relevant cost.

The Kislaya Guba TPP is equipped with an electrolysis unit to produce chlorine, which is used to keep aquatic microorganisms away from the waterway during breeding season. An investigation in 1998 stated

Overview over available reversible pump-turbines (RPT)

	Axial flow pumps used as turbine (PAT)	Mixed flow PATs	Radial pump turbines	Archimedes screw	Positive displacement pump-turbine
Operative head range (m)	1–5 [94]	5–15 [94]	>60 [95]	2-10 [96-98]	-
Characteristics	Flow rates up to 1000 l/s. [94]	Flow rates of 50–150 l/s. [94]	Power usually exceeding 50 MW [95]	Discharge ranges up to 15 m <sup>3</sup> /s. Low installation and maintenance costs, in turbine mode has been used with efficiencies up to 90% [99,100], but efficiency declines steeply with changes in water level [101]. Compared to conventional bladed pump-turbines, they are more fish friendly [97,99].	Low specific speed [102]. Tested as micro hydro turbines with pressures up to 5 bar (51 m) having efficiencies between 60%–80% [103,104]. Given their low speeds, they can be regarded as fish friendly [105]

that the electrolysis unit protected the waterway and the turbine from fouling for 20 years [85]. Additionally, the experience from Kinslaya shows that chlorine was not detected outside of the penstock [85].

However, according to the European Chemical Agency (ECHA), chlorine could be a threat to surrounding flora and fauna [86]. Furthermore, experience from Annapolis shows that the use of chloride in seawater caused deposition of manganese, which had to be scraped out every three to four years [87]. Nowadays the focus is on alternatives such as electrochemical degradation that minimize harm to the surrounding environment. This technology uses electric fields to kill fouling organisms directly, i.e. without producing a biocidal intermediate (such as chlorine) [88]. Furthermore, following the ban of environmentally harmful tributyltin (TBT)-based paints, the need for alternative technologies with comparable effectiveness and a lack of negative effects on aquatic flora and fauna is elevated.

In the case of SPHS, it is possible to hypothesize a future application of next generation antifouling technologies based on the use of natural biocides as additives to coatings [89,90], and other strategies such as the application of non-stick fouling release coatings [91] based on ultra-smooth surfaces.

# 5. Pump turbines

Conventional mountainous high-head PHS can operate at round trip efficiencies between 70% and 85% [92]. However, the pump-turbine units of these plants are not efficient for low-head applications (see Fig. 2). The available power (W) follows the following equation:

$$P = \rho \cdot g \cdot H \cdot Q \cdot \eta \tag{4}$$

where  $\rho$  is the density of the water (kg/m<sup>3</sup>), g gravity acceleration (m/s<sup>2</sup>), H head difference between basins (m), Q volumetric flow rate and (m<sup>3</sup>/s),  $\eta$  the overall efficiency of a power plant (-). Eq. (4) shows that for low-heads, the volumetric flow rate must be increased to produce high power [93]. To limit system loss, for large discharges the diameter of the penstocks and pump–turbine runners must also be large (to keep flow velocity low).

As seen in Fig. 2, different kinds of turbines are optimized for different head and discharge ranges. In Table 7, different pump–turbines types suitable for low-head applications are shown.

Recently counter-rotating pump-turbines have been considered a promising technology for low-head PHS. Compared to conventional pump-turbines, the counter-rotating type can be of smaller size and have higher efficiency for a larger head range [106].

Having pump-turbines able to work at variable rotational speeds, allows for high efficiency and more power over a larger range of

heads [107,108]. However, the costs of variable speed machines are around 30% higher than for fixed speed machines. Thus, the choice between each option relies on techno-economic and demand, [108,109] even though variable speed units ensure a higher renewable energy penetration [109].

For more information on pump–turbine preferences as well as other technological aspects (grid integration and electric machines and control) regarding low-head SPHS the reader is referred to Hoffstaedt et al.. 2022 [110].

# 6. Legal and environmental aspects

# 6.1. Legal aspects

As with any infrastructure project, PHS projects face legal and permitting tasks, whether on land or in the sea. To investigate the issues involved, we take as an example a hypothetical low-head PHS project located within waters subject to German administrative law, keeping in mind that other projects around the North or Baltic seas under other jurisdictions might face similar requirements, as the overall principles impacting the different national administrative laws are often similar.<sup>1</sup> Under the assumption that the plant will be entirely placed in the sea, the 1982 United Nations Convention on the Law of the Sea (UNCLOS) provides for – inter alia – two different regimes with different legal prerequisites:

- 1. The Exclusive Economic Zone ("EEZ ") beyond the Territorial Sea and up to 200 nautical miles from the baseline (coast). This area does not belong to the adjacent state, which nevertheless can exercise certain rights regarding the use of this part of the sea and its natural resources. This zone plays an increasingly important role regarding the installation of electricity generating plants, as offshore windfarms nowadays are mostly installed farther away from the coastal area.
- 2. The Territorial Sea up to 12 nautical miles from the coast. In this zone, the adjacent state has full sovereignty.

Within the German EEZ in the North Sea or Baltic Sea, projects require permits from the Federal Maritime and Hydrographic Agency (Bundesamt für Seeschifffahrt und Hydrographie, BSH), which enacts a single integrated permitting procedure that includes publication of planning documents and participation of all affected public and private

<sup>&</sup>lt;sup>1</sup> This chapter does not constitute and shall not be considered as specific legal advice.

stakeholders. The relevant law for construction in the EEZ is the Act governing plants in the EEA (SeeAnIG) [111]. Roughly summarized, the project permit might be granted if such construction does not endanger the maritime environment. Examples of harmful effects are seawater pollution, conflict with birds' migration routes, conflict with marine transport lanes, or conflicts with existing/ projected cables or converter stations (offshore windfarms). For the special case of a power plant not connected to the public power network or in case that the competent authority has set up special areas for "other energy generating plants", the Act regarding wind energy in the sea (WindSeeG) [112] providing for a tender procedure might apply. In either case, the crucial question will be to what extent the project will have a (negative) impact on the environment. Project owners will have to provide the authority with either a formalized environmental impact assessment ("EIA") according to the Federal Act governing the environmental impact assessment (UVPG) [113], or at least with an expert assessment regarding the potential impact on environmental aspects protected by the SeeAnlG.

Within the German EEZ, there are "Natura 2000" areas. The Council Directive on the conservation of natural habitats and of wild fauna and flora [114] and the Directive on the conservation of wild birds [115] was transferred into national law by integrating its provisions into the Act regarding the protection of the environment (BNatSchG) [116]. Generally, a PHS plant in the "Natura 2000" areas, that fall under EU legislation, might only be authorized if the applicant can demonstrate (by expert opinions) that there will not be any severe negative impacts on the "Natura 2000" protected areas, a burden which probably will be rather high.

Unlike for the German EEZ, construction in the German territorial sea is subject to the adjacent German federal state as the competent authority to grant the construction and operating permits. As for projects in the EEZ, the permitting process necessitates the involvement of all stakeholders and the consultation of all other authorities being the competent authorities for issues which might potentially be affected by the project. The relevant law is the Federal Water Resources Act (WHG) [117]. According to this Act the construction of a plant such as a PHS plant will be considered as modification of a waterbody. The applicant has to provide the authority with an environmental impact assessment according to UVPG. It should be noted that the use of the seawater and its "recycling" back into the sea are as well subject to permits according to the WHG. Within the territorial seas, special attention must be paid to the fact that according to the BNatSchG there are 3 national parks in the German coastal zone. The applicable laws regarding these national parks stipulate to which extent the "use" of these national parks will be allowed. Generally, the laws restrict or even ban any use which might endanger the purpose of the protection as national parks.

Though this section mainly deals with the specific case of German administrative law as relates to permitting of a PHS facility in the German EEZ or territorial waters, similar procedures exist throughout the EU. Such procedures will need to be followed in order for projects to be permitted.

However, national laws might be modified in the course of the energy transition process taking into account new technologies with wider acceptance as environmentally friendly and economic methods to achieve long-term climate targets.

# 6.2. Environmental aspects

Coastal marine areas are characterized by the highest values of ecosystem services and by multiple uses that are often in conflict with one another [118]. Hydropower applications pose a series of environmental impacts: some do not differ much from those caused by coastal infrastructural works (e.g. ports, coastal works), others are technology-specific but mostly result in fish mortality, either directly or indirectly [119]. Ecologically-sensitive areas require careful planning and thorough assessments of the ecosystem impacts. Hydrodynamics and seabed morphology are the two most affected compartments along all construction phases in turn affecting biodiversity and ecosystem functioning [50, 120]. In particular, (i) changes in salinity and oxygen stratification reflecting on organic matter decomposition rates, (ii) increased turbidity of coastal water affecting photosynthetic efficiency of primary producers, (iii) disrupted transport of fish larval stages posing threats to nursery areas and (iv) an overall impoverishment of the habitat structural complexity threatening pelagic, benthic and avian species were reported.

Mitigation measures, defined as "measures that would avoid, reduce and, if possible, offset significant adverse effects", are outlined in the Environmental Impact Assessment report produced at the end of the EIA process. Impacts can be minimized by scaling down or relocating the project site as well as by applying additional measures to reduce the impacts either at the source or at the receptor [121].

Offset measures compensate for negative biodiversity impacts by providing measurable conservation gains that persist at least as long as projected impacts [122]. Offsets may consist in management interventions such as restoration of degraded habitats or spatial extension of already protected sites such as Special Areas of Conservation (SACs), Special Protection Areas (SPAs) and Site of Community Importance (SCIs) to counteract the loss of biodiversity, habitats or species. Specifically, restoration measures can be classified as active, when habitat-forming species are transplanted/restocked (e.g. Piazzi, 1998 [123]), or passive, achieved with the use of both artificial (e.g. artificial reefs) [124] and natural (e.g. honeycomb worm *Sabellaria alveolata*) hard substrates [125].

Enhancement measures improve the overall biodiversity levels over time in that so-called umbrella species provide adequate structural complexity for the ecosystem. Their selection must be based on precise criteria [126]. The European flat oyster (*Ostrea edulis*) is an optimal candidate because their biogenic structures facilitate the resettlement of several associated benthic species and can attenuate waves and retain sediments [53].

The European legal requirements impose a good ecological status (GES) of marine waters that is to be achieved by protecting biological communities [127]. The status of fish populations, which are considered by multiple descriptors among the 11 of the Marine Strategy Framework Directive, is first investigated in conditions of minimal anthropogenic impact pre-installation (baseline); hydropower-related impacts are then assessed in terms of survival, with particular reference to migration routes and spawning grounds: mathematical modeling predicts turbine encounter and fish injury rates, which are validated by netting/electrofishing seasonal surveys; cameras and sensors assess fish behavior in the vicinity of operating turbines and record the flow conditions; radio tagging and recapture techniques at experimental sites quantify the extent of physical injuries for developing exact mitigation measures; acoustic telemetry reconstructs the 2D behavioral preferences and migration pathways of fish at dams which are then exploited to design optimal passage solutions [128,129].

Hydropower impacts on local and migratory fish may be primarily mitigated by designing environmentally friendly turbines [130, 131], installing fish protection screens and constructing fishways allowing for the bypass of hydroelectric plants or dams. Prompted by various stakeholders (e.g. WWF Switzerland and European Rivers Network coordinating the Salmon Come Back campaign [132]; Association Saumon-Rhin [133]), both academic research and environmental consulting companies assess fish passability efficiency, a subject that has received increasing attention in the last two decades, by characterizing fish population structure and genetics. Should the structure fail to perform as intended, power companies may opt for solutions as drastic as to demolish power stations, as was the case for the EDF Vogelgrün plant in 2017. Hydropower technologies have so far been evaluated as fishfriendly based on the severity of injuries (e.g. scale loss, haemorrhages, skin wounds) and mortality rates caused by the system physical processes [134], even though precise and universally-descriptive statistics cannot be provided due to the many factors, e.g. type of damage, fish species, life stages, hydropower characteristics, that must be kept in consideration. On the other hand, the study of animals' physiological status has been entirely overlooked. Yet, fish, as all vertebrates, do respond to stressful stimuli such as confinement, handling or physical disturbance [135], and stress levels are paramount for animal welfare. Severe and/or sustained stress responses have neuromodulatory and behavioral effects with major life history consequences [136,137]; these reflect on higher biological levels in terms of inhibitory or interruptive effects on growth, reproduction, behavior and disease resistance [138]. The concentration of the cortisol hormone, a widely recognized stress indicator, can be monitored as a proxy of acute stress.

# 7. Costs and benefits

# 7.1. Tidal plants

The high capital cost of TPP construction is the main barrier against more wide-spread development and serves as a valuable lesson for potential low-head SPHS. Capital costs are site specific and vary with the geography and hydrology of a site [139]. The total project costs of La Rance TPP are  $\in$ 801 million today [140] ( $\in$ 3340/kW). However, the dam and the powerhouse construction of the Sihwa TPP generated costs discounted to today of  $\in$ 689 million [141] and  $\in$ 332 million [23] respectively ( $\in$ 4020/kW). Compared to the global weighted-average total installed costs of hydropower projects,  $\in$ 1263/kW in 2018 according to IRENA [142], capital costs of TPPs are considered to be high.

Tidal barrage power plants such as La Rance and Sihwa have been found to have a levelized cost of energy (LCOE) on the same order as the  $\notin$ 70/MWh that represents the global average for hydropower plants in general [143–145].

Various equations to predict the cost of energy and the costeffectiveness of sites are available. Baker [146] approximates the tidal unit cost of electricity (p/kWh) empirically. This author uses physical dimensions like the length of the barrage, the area of the basin and the mean tidal range [22]. Another method for estimating the cost of energy generated by a TPP is by adding the capital recovery cost, which takes into account the amortization period, and operation and maintenance costs per unit of energy [139]. The Gibrat ratio [147] can be used to estimate the cost-effectiveness of a TPP site. It compares the length of the barrage in meters to the annual energy production in kWh. Therefore, the most profitable site has the smallest Gibrat ratio.

# 7.2. Pumped hydro storage plants

PHS's prime purpose is balancing energy supply and demand on a daily basis (arbitrage) and is secondary purpose is to offer black start and renewable integration capacity. It is expected that by 2040, hydrogen will be the preferred technology for weekly and seasonal storage, PHS for daily storage, and Li-ion batteries for grid balancing from timescales of minutes to hours [148]. Since low-head PHS relies on similar technology as traditional high-head PHS it would be most competitive when equipped with a storage capacity stretching from 10 to 30 h. This is the same range as in the literature [67,149]. Besides daily balancing, a secondary revenue stream can consist of black start support and ancillary services [150].

The dam construction, the pump-turbines and the powerhouse and conveyance system each account for approximately  $\frac{1}{3}$  of the total costs [50,67,120,149]. The costs of the dam are highly dependent on the required bank revetment, which can form half of the total dam costs in rough sea conditions [50]. For the pump-turbines it is beneficial to use the largest installable runner possible in order to minimize the number of pump-turbines and their corresponding electrical infrastructure [151,152]. A factor that can limit the runner diameter is the

constructability of the conveyance system. Cavitation prevention and the use of draft tubes for a vertical axis pump-turbines can lead to larger construction depths and greater initial expense.

The costs of a circular reservoir with a theoretical vertical dam scale with the storage capacity to the power 0.5 (dam costs rise linearly with the diameter, and the surface area quadratic). For a dam that is not vertical but has an inner slope (e.g. 1:5) the scaling benefit increases, since a larger reservoir 'loses' relatively less storage volume due to the dam structure. This leads to a scaling power of the storage capacity costs of approximately 0.4 [50].

For the power capacity, repeated construction of powerhouse units and conveyance systems to house multiple turbines will result in a scale advantage [152], due to the learning curve during construction, economies of scale and the sharing of equipment and mobility costs.

Maintenance and operational costs for standard PHS vary from 1% to 2.2% of the CAPEX per year [153,154]. To date, the additional expenses to protect and maintain a PHS facility against the saline environment are unspecified, though it is estimated that the la Rance tidal plant incurs annual operation and maintenance (O&M) costs of 1.5% to 2% of its capital cost (EDF, personal communication, 2021). Experiences from the Okinawa Yanbaru seawater PHS station and the La Rance tidal barrage do show that it is possible to operate successfully in corrosive salt water conditions [155], (EDF, personal communication, 2021).

For comparing energy storage technologies and their competitiveness, the levelized cost of storage (LCOS) is generally used. The LCOS includes i.e. the investment costs, annual operational costs, interest rate, lifetime and the yearly 'produced' amount of electricity. New PHS plans are estimated to have an LCOS ranging from  $\in$ 50/MWh to  $\in$ 80/MWh, while Lithium-ion battery prices are expected to fall to  $\in$ 100/MWh by 2040 [156]. Note that even with the LCOS it is difficult to make a fair comparison between large scale and small scale technologies, hence Lithium-ion batteries are still expected to be the preferred technology for short-term and small-scale storage services [148].

For low-head seawater PHS the LCOS is expected to vary between €40/MWh and €140/MWh for a best case and worst case scenario respectively. From this range €100/MWh forms the basic scenario.<sup>2</sup> The large range in LCOS is mainly caused by uncertainty regarding the operational and maintenance costs, the amount of turbining hours per day and the location of the storage system and the corresponding deferral of investments in transmission capacity that can be taken into account. Costs for transmission infrastructure are not included, but will benefit technologies that can be applied locally next to the energy source, like a SPHS scheme next to an offshore wind farm. For a storage basin with 2.65 GW of installed power in the middle of the North Sea the grid deferral can save up to €30/MWh when discounted to the LCOS [51]. This potential saving also applies to any other storage technique that can be applied locally, but not to e.g. PHS, for which the energy first has to be transmitted to a basin in a mountainous region. The LCOS analysis indicates that low-head seawater PHS can be a competitive technology to integrate large shares of renewable energy into the grid.

<sup>&</sup>lt;sup>2</sup> For the LCOS basic scenario the capital and operational costs from De Vilder (2017) were used for an offshore facility with 25 GWh of storage capacity and 2.65 GW of turbine power in 20 m deep water and protected against 8 m high waves all around. The capital costs are estimated at €4.65 billion with €110 million per year of operational and maintenance costs. Furthermore 6 turbining hours per day are assumed, 70% roundtrip efficiency, 2% downtime, 5% interest rate and a 100 year lifetime with two complete refurbishments of all the electrical and mechanical parts.

# 8. Discussion

Most of the experience in seawater energy generation techniques originates from TPP. The only executed SPHS technology remains the Okinawa plant, yet several other projects are expected to be realized in the near future such as the Espejo de tarapacá SPHS. Respecting lowhead PHS, no plan has been realized yet. The novelty of the low-head PHS technology opens several research fields tackled in this review paper such as new reversible pump-turbine technology, maintenance of electrical and mechanical components in seawater, methods for identifying adequate locations, ecological and morphological effects as well as the construction of innovative structures and dealing with legislation of multiple countries and localities, thus presenting multiple research opportunities.

Equipment placed in a marine environment suffers more corrosion compared to freshwater. Additionally, equipment in seawater accumulates microorganisms (fouling) that can deteriorate the equipment and reduce efficiency. Several techniques have been discussed that have been put into practice in existing TPPs and in the Okinaka SPHS plant. Effective corrosion prevention measures are cathodic protection, the use of corrosion-resisting materials (stainless steel, FRP, GRP) and the use of coating paint. Antifouling measures include the use of natural biocides in coatings as well as non-stick fouling release coatings. Most TPPs were constructed in the second half of the 20th century, thus including antifouling and anti-corrosion techniques from that time. This review shows that nowadays there is potential for research into new environmentally friendly antifouling and anti-corrosion measures.

As an example of legal constraints to SPHS, a review of the German law regarding construction in the coastal zone was conducted. In case that a suitable location for low-head SPHS is found in a Natura 2000 area, the construction of such a facility will only be approved if it is shown that the project brings more ecological gain in the long term than the damage it causes to the protected area. The novelty of low-head SPHS technologies creates uncertainty in how the marine environment will react to such a project. Due to such legal and environmental constraints, the iLand project already encountered some delays in project execution due to the multiple jurisdictions it was subject to. This paper recognizes legal and environmental issues as critical showstoppers together with investment costs.

Floating solar, which has been shown to produce a mutual benefit with hydropower reservoirs by reducing evaporation while also cooling and enhancing the efficiency of the solar operation, could provide the same mutual benefit with PHS

# 9. Conclusion

For low-head PHS, a high reservoir is recommended in shallow water (5 to 10 m) and in deeper water (10 to 30 m) a valmeer alternative is more beneficial when looking at the volume balance between dredging and dam construction. However, a high reservoir in shallow (coastal) water can require a higher safety level, due to an increased flood risk to surrounding areas. Even though a higher safety level would imply higher expenses, it is not expected that it would be more costly than a valmeer alternative with a largely unequal dredged volume balance.

Failure mechanisms such as soil uplift, piping and macro-instability due to rapid drawdown have been identified from literature as the most relevant regarding the dam structure for low-head PHS. When adequately designed, these failure mechanisms are not technical showstoppers.

TPPs are currently a competitive energy generating alternative, demonstrating that seawater hydroelectricity can be economically feasible. Similarly to conventional PHS, low-head SPHS would be most competitive when the timescale of storage is between 10 and 30 h. Maintenance costs in seawater are higher than in freshwater but experiences from TPPs and the Okinawa Yanbaru SPHS plant show that operation in this environment is feasible. Finally, this paper shows that low-head SPHS LCOS (levelized costs of storage) vary between  $\notin$ 40/MWh and  $\notin$ 140/MWh. Thus it is shown that low-head SPHS can be a competitive technology for integrating variable renewable energies into the grid.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Acknowledgments

Bettina Geisseler, GEISSELER LAW - The legal aspects chapter (Chapt. 6.1) could not have been written without the support/information by:

- Mr. Klaus Gossen, Niedersächsischer Landesbetrieb für Wasserwirtschaft, Küsten- und Naturschutz (NLWKN) - Direktion
- Dr. Andreas Wurpts, Leiter Forschungsstelle Küste/Coastal Research Station, NLWKN
- Dr. Nico Nolte, BSH

We thank Voith GmbH for allowing us to use their figure "application range of Voith Hydro's turbines" (Fig. 2).

This paper was funded by the European Union Horizon 2020 project ALPHEUS (Augmenting grid stability through Low-head Pumped Hydro Energy Utilization & Storage) project ID 883553.

# References

- European Comission. A European green deal. 2021, European Commission -European Commission.
- [2] Eurostat. Shares of energy from renewable sources. 2018, European Comission.
- [3] Bird Lori, Lew Debra, Milligan Michael, Carlini E Maria, Estanqueiro Ana, Flynn Damian, et al. Wind and solar energy curtailment: A review of international experience. Renew Sustain Energy Rev 2016;65:577–86.
- [4] TWENTIES. Transmission system operation with a large penetration of wind and other renewable electricity sources in electricity networks using innovative tools and integrated energy solutions. TWENTIES final report. 2013.
- [5] Burke Daniel J, O'Malley Mark J. Factors influencing wind energy curtailment. IEEE Trans Sustain Energy 2011;2(2):185–93.
- [6] Matevosyan Julija. Wind power integration in power systems with transmission bottlenecks. In: 2007 IEEE power engineering society general meeting. 2007, p. 1–7.
- [7] Moradzadeh Mohammad, Zwaenepoel Brecht, Van de Vyver Jan, Vandevelde Lieven. Congestion-induced wind curtailment mitigation using energy storage. In: 2014 IEEE international energy conference. 2014, p. 572–6.
- [8] Zou Jin, Rahman Saifur, Lai Xu. Mitigation of wind output curtailment by coordinating with pumped storage and increasing transmission capacity. In: 2015 IEEE power energy society general meeting. 2015, p. 1–5.
- [9] Ma Tao, Yang Hongxing, Lu Lin. Feasibility study and economic analysis of pumped hydro storage and battery storage for a renewable energy powered island. Energy Convers Manage 2014;79:387–97.
- [10] Katsaprakakis Dimitris Al, Christakis Dimitris G. Seawater pumped storage systems and offshore wind parks in islands with low onshore wind potential. a fundamental case study. Energy 2014;66:470–86.
- [11] Ioakimidis Christos S, Genikomsakis Konstantinos N. Integration of seawater pumped-storage in the energy system of the island of São Miguel (Azores). Sustainability 2018;10(10).
- [12] Pradhan Anish, Marence Miroslav, Franca Mário J. The adoption of seawater pump storage hydropower systems increases the share of renewable energy production in small island developing states. Renew Energy 2021;177:448–60.
- [13] Energy Australia. Cultana pumped hydro project: ARENA final report. 2017.
- [14] International Hydropower Association (IHA). Hydropower status report: sector trends and insights. Technical report, 2021.
- [15] Chen Haisheng, Cong Thang Ngoc, Yang Wei, Tan Chunqing, Li Yongliang, Ding Yulong. Progress in electrical energy storage system: A critical review. Prog Nat Sci 2009;19(3):291–312.
- [16] Barnhart Charles J, Benson Sally M. On the importance of reducing the energetic and material demands of electrical energy storage. Energy Environ Sci 2013;6(4):1083–92.

- [17] Nikolaidis Pavlos, Poullikkas Andreas. Cost metrics of electrical energy storage technologies in potential power system operations. Sustain Energy Technol Assess 2018;25:43–59.
- [18] Ansorena Ruiz Ruben. Conceptual design of the Valmeer's pump storage station of the DELTA21 plan. 2020, Technische Universiteit Delft.
- [19] Vincent de Laleu ; EDF. BHA annual conference liverpool. 2009.
- [20] Bernshtein Lev B. Tidal power development-a realistic, justifiable and topical problem of today. IEEE Trans Energy Convers 1995;10(3):591–9.
- [21] Li Ying, Pan Dong-Zi. The ebb and flow of tidal barrage development in Zhejiang province, China. Renew Sustain Energy Rev 2017;80:380–9.
- [22] Hammons Thomas James. Tidal power. Proc IEEE 1993;81(3):419–33.
- [23] Bae Young Ho, Kim Kyeong Ok, Choi Byung Ho. Lake sihwa tidal power plant project. Ocean Eng 2010;37(5–6):454–63.
- [24] Neill Simon P, Angeloudis Athanasios, Robins Peter E, Walkington Ian, Ward Sophie L, Masters Ian, et al. Tidal range energy resource and optimization – past perspectives and future challenges. Renew Energy 2018;127:763–78.
- [25] Delory RP, Eng P. The annapolis tidal generating station. Water for energy: papers presented at the 3rd international symposium on wave, Tidal, OTEC, and small scale hydro energy. 1986, Brighton, England, 14 - 16 May, 1986:125 - 132. 1986.
- [26] Bernshtein LB, Gavrilov VG, Gel'fer SL, Nehoroshev UN, Suponitskii LI, et al. Fisheries and marine service - translation series no. 4078 - the kislaya guba tidal power plant. Department of Fisheries and the Environment Fisheries and Marine Service Scientific Information and Publications Branch Ottawa, Ont. 1977.
- [27] International Hydropower Association (IHA). Factsheets: Pumped storage hydropower. 2021.
- [28] Gougenheim Andrè. L'utilisation de l'ènergie des marès. Vol. 83. Ciel Et Terre; 1967, p. 173.
- [29] Bernshtein LB. Problem of the utilization of tidal energy and the experimental Kislaya Guba tidal powder plant. Hydrotech Constr 1969;3(1):12–8.
- [30] Rice R, Baker G. The annapolis experience. In: OCEANS'87, IEEE; 1987, p. 391–6.
- [31] Koh Chul-Hwan. Korean megatidal environments and tidal power projects: Korean tidal flats-biology, ecology and land uses by reclamations and other feasibilities. Houille Blanche 1997;(3):66–78.
- [32] Electric Power Development Co, Ltd. (Translated from Japanese) about the start of demonstration test operation of okinawa yanbaru seawater pumping power station. 1999.
- [33] Japan Update. Experimental power plant in Kunigami dismantled. 2016, Japan Update.
- [34] Katsaprakakis Dimitris Al, Christakis Dimitris G, Stefanakis Ioannis, Spanos Petros, Stefanakis Nikos. Technical details regarding the design, the construction and the operation of seawater pumped storage systems. Energy 2013;55:619–30.
- [35] Katsaprakakis Dimitris Al, Voumvoulakis Manolis. A hybrid power plant towards 100 sifnos, Greece. Perspectives created from energy cooperatives. Energy 2018;161:680–98.
- [36] Gestión Ambiental Consultores (GAC). Capítulo 1: descripción de proyecto EIA espejo de tarapacá. Technical report, 2014.
- [37] Organic Power Ltd. Energy storage hub: glinsk mountain, county mayo (op18 project brochure issue 3). Technical report, Organic Power Ltd; 2011.
- [38] Water Power Magazine. At the hub of it all. 2012, https://www. waterpowermagazine.com/features/featureat-the-hub-of-it-all/.
- [39] NS Energy Business Magazine. Cultana pumped hydropower project, port Augusta, Australia. 2021.
- [40] ARUP and Energy Australia. CultanaPumped hydro energy storage project, phase 2. ARENA knowledge sharing report (Public). techreport, Australian Renewable Energy Agency (ARENA); 2020.
- [41] Peña Karen. Valhalla ajusta fechas de sus proyectos eléctricos y continúa en búsqueda de financiamiento. 2021, Diario Financiero.
- [42] NUnl online newspaper. Plan Lievense: al in 1981 werd er nagedacht over opslag van groene stroom. 2019, NU.
- [43] DE INGENIEUR. Lievense, de man van het opslagbekken. 2015, De Ingenieur.
- [44] Rijkswaterstaat (RWS), Hollandsche Beton Groep NV, Ballast Nedam Groep NV, RaadgIngBurLievense. Pomp accumulatie centrale Brouwersdam en IJsselmeer (Hoofdrapportage fase 2). Technical report, Rijkswaterstaat; 1986.
- [45] de Boer WW, Verheij FJ, Moldovan N, Van der Veen W, Groeman F, Schrijner M, et al. Energie-eiland, de haalbaarheid van drie verschillende opties van energieopslag voor Nederland. Technical report, KEMA Consulting/Lievense; 2007.
- [46] Libaux A, Grant C, Burrows R, Walkington IA. Tidal power plant design and optimization. Sacramento, CA: Hydrovision International; 2011.
- [47] Bernshtein LB. Tidal power plants; Korea ocean research and development institute: Seoul. 1996, Korea.
- [48] European association for the cooperation of transmission system operators (TSOs) for electricity (ENTSO-E). Project 1002 - iland. 2021.
- [49] Lavooij Huub, Berke Leen. UPDATE DELTA21 29-7-2019. 2019.
- [50] de Vilder Lucas. Offshore pumped hydropower storage: Technical feasibility study on a large energy storage facility on the dogger bank. 2017, Technische Universiteit Delft.

- [51] Gerrits Stefan. Feasibility study of the hub and spoke concept in the north sea: Developing a site selection model to determine the optimal location. 2017.
- [52] van Adrichem Steve. Influence of rapid draw down on dike stability. 2021, Technische Universiteit Delft.
- [53] Frölke G. Nature enhancing opportunities for coastal defences: Opportunity study for the dogger bank. 2017, Technische Universiteit Delft.
- [54] Kibrit B. Pumped hydropower storage in the netherlands. 2013, Technische Universiteit Delft.
- [55] Onwuachu PRI. The new haringvliet barrier: Conceptual design for the storm surge barrier of the Delta21 project. 2021, Technische Universiteit Delft.
- [56] Voith GmbH's webpage. Turbines & shut-off valves. 2021.[57] Wang ZJ, Wang ZW. A review on tidal power utilization and operation
- optimization. IOP Conf Ser Earth Environ Sci 2019;240(5):052015.
- [58] Gordon Tidmarsh W. Assessing the environmental impact of the annapolis tidal power project. Water Sci Technol 1984;16(1–2):307–17.
- [59] Chang J, Leung Dennis YC, Wu CZ, Yuan ZH. A review on the energy production, consumption, and prospect of renewable energy in China. Renew Sustain Energy Rev 2003;7(5):453–68.
- [60] Charlier Roger H, Ménanteau Loïc, Chaineux Marie-Claire P. The rise and fall of the tide mill. 2004.
- [61] Novak Pavel. Developments in hydraulic engineering. CRC Press; 1987.
- [62] Rourke Fergal O, Boyle Fergal, Reynolds Anthony. Tidal energy update 2009. Appl Energy 2010;87(2):398–409.
- [63] Bernshtein LB, Usachev IN. Utilization of tidal power in Russia in overcoming the global energy and ecological crisis. Houille Blanche 1997;(3):96–100.
- [64] Duhoux L. Comportement en service des principaux ouvrages de génie civil de l'usine de la rance. Houille Blanche 1973;(2–3):153–62.
- [65] Mondt W. Het energieatol energieopslag in de Noordzee [Presentation slides]. 2013.
- [66] Monclus Abadal Albert, Dornhelm Esther, Mohamed Khaled Aly Mohamed Elakel Mohamed, Feys Charles, Naderi Nader. Energy hub in the north sea. 2017.
- [67] Rijkswaterstaat (RWS), Hollandsche Beton Groep NV, Ballast Nedam Groep NV, RaadgIngBurLievense. Pomp accumulatie centrale Brouwersdam en IJsselmeer (Deelrapportage fase 2 Veiligheid). Technical report, Rijkswaterstaat; 1986.
- [68] Helpdesk Water(nd). Normering. Waterveiligheidsportaal. 2022.
- [69] Battjes JA, Labeur RJ. Open channel flow. TU Delft, Department Hydraulic Engineering; 2014, Collegedictaat CTB3350.
- [70] Winsemann Jutta, Alho Petteri, Laamanen Leena, Goseberg Nils, Lang Jörg, Klostermann Josef. Flow dynamics, sedimentation and erosion of glacial lake outburst floods along the middle pleistocene scandinavian ice sheet (northern central europe). Boreas 2016;45(2):260–83.
- [71] Rijkswaterstaat (RWS), Hollandsche Beton Groep NV, Ballast Nedam Groep NV, RaadgIngBurLievense. Pomp accumulatie centrale Noordzeekust (Deelrapportage fase 1 Bekkendijken). Technical report, Rijkswaterstaat; 1985.
- [72] Van Dam JL. Ontwerp duinenrij energie-opslagmeer. 2020, Haagse Hogeschool.
- [73] L'usine marémotrice de la rance. Houille Blanche 1962;(2):117–29.
- [74] Bernshtein LB, Usachev IN. The kislaya guba tidal power plant. 1972, p. 347.[75] Cha Hung Youn, Kim Ki Cheol, Ahn Jeong Hwan, Jang Dae Woon. Information
- of main facilities of sihwa tidal power plant. Mag Korea Inst Struct Maint Insp 2012;16(3):4-9.
- [76] Clark Robert H. Elements of tidal-electric engineering. Vol. 33. John Wiley & Sons; 2007.
- [77] Bernshtein LB. The journal's role in disseminating information on certain promising designs of hydroelectric and tidal power plants. Hydrotech Constr 1980;14(11):1168–70.
- [78] Paasman Yordi. Design for the in- and outlet structure of the energy storage lake within the Delta21 plan. 2020, Technische Universiteit Delft.
- [79] Lambert M. Bilan de la protection cathodique des ouvrages du barrage de la rance. de la conception au résultat final. Houille Blanche 1997;(3):36–41.
- [80] Fujihara Tetsuo, Imano Haruo, Oshima Katsuhiro. Development of pump turbine for seawater pumped-storage power plant. Hitachi Rev 1998;47(5):199–202.
- [81] Yebra Diego Meseguer, Kiil Søren, Dam-Johansen Kim. Antifouling technology past, present and future steps towards efficient and environmentally friendly antifouling coatings. Prog Org Coat 2004;50(2):75–104.
- [82] Cattò Cristina, Cappitelli Francesca. Testing anti-biofilm polymeric surfaces: Where to start? Int J Mol Sci 2019;20(15):3794.
- [83] Champ Michael A. A review of organotin regulatory strategies, pending actions, related costs and benefits. Sci Total Environ 2000;258(1–2):21–71.
- [84] Abbott A, Abel PD, Arnold DW, Milne A. Cost-benefit analysis of the use of TBT: the case for a treatment approach. Sci Total Environ 2000;258(1–2):5–19.
- [85] Usachev IN. Results of 30-year monitoring of the marine concretes and floating structure of the kislogubsk tidal electric power plant in the far north. Hydrotech Constr 1998;32(12):705–12.
- [86] European Cheminals Agency (ECHA). Chlorine substance infocard. 2021.
- [87] Clark RH. Prospects for fundy tidal power. Houille Blanche 1997;83(3):79-85.
- [88] Chen X, Noy A. Antifouling strategies for protecting bioelectronic devices. APL Mater 2021;9(2).
- [89] Abarzua Sibylle, Jakubowski Sabiene. Biotechnological investigation for the prevention of biofouling. I. biological and biochemical principles for the prevention of biofouling. Mar Ecol Prog Ser 1995;123:301–12.

- [90] Rittschof Dan. Natural product antifoulants: one perspective on the challenges related to coatings development. Biofouling 2000;15(1–3):119–27.
- [91] CEPE (Conseil Européen de l'Industrie des Peintures, des Encres d'Imprimerie et des Couleurs d'Art) Antifouling Working Group. Utilisation of more 'environmental friendly' antifouling products - phase 1 final report. Technical report, EC project No 96/559/3040/DEB/E2, 1999.
- [92] Rehman Shafiqur, Al-Hadhrami Luai M, Alam Md Mahbub. Pumped hydro energy storage system: A technological review. Renew Sustain Energy Rev 2015;44:586–98.
- [93] Hunt Julian David, Byers Edward, Riahi Keywan, Langan Simon. Comparison between seasonal pumped-storage and conventional reservoir dams from the water, energy and land nexus perspective. Energy Convers Manage 2018;166:385–401.
- [94] Carravetta Armando, Houreh Shahram Derakhshan, Ramos Helena M. Pumps as turbines: Fundamentals and applications. In: Springer tracts in mechanical engineering, Springer International Publishing; 2018.
- [95] Bogenrieder W. 2.6. Pumped Storage power plants. In: Renewable energy. Springer; 2006, p. 165–96.
- [96] Okot David Kilama. Review of small hydropower technology. Renew Sustain Energy Rev 2013;26:515–20.
- [97] Ak Mümtaz, Kentel Elçin, Kucukali Serhat. A fuzzy logic tool to evaluate lowhead hydropower technologies at the outlet of wastewater treatment plants. Renew Sustain Energy Rev 2017;68:727–37.
- [98] YoosefDoost Arash, Lubitz William David. Archimedes screw turbines: A sustainable development solution for green and renewable energy generation—A review of potential and design procedures. Sustainability 2020;12(18):7352.
- [99] Zhou Daqing, Deng Zhiqun Daniel. Ultra-low-head hydroelectric technology: A review. Renew Sustain Energy Rev 2017;78:23–30.
- [100] Waters Shaun, Aggidis George A. Over 2000 years in review: Revival of the archimedes screw from pump to turbine. Renew Sustain Energy Rev 2015;51:497–505.
- [101] B.V. Spaans Babcock. APPENDIX 2 archimedean screw pump details. Technical report, Spaans Babcock B.V.; 2018.
- [102] Karassik Igor J, Messina Joseph P, Cooper Paul, Heald Charles C. Pump handbook. McGraw-Hill Education; 2008.
- [103] Kurokawa Junichi, Matsui Jun, Choi Young-Do. Flow analysis in positive displacement micro-hydro turbine and development of low pulsation turbine. Int J Fluid Mach Syst 2008;1(1):76–85.
- [104] Sonawat Arihant, Choi Young-Seok, Kim Kyung Min, Kim Jin-Hyuk. Parametric study on the effect of inlet and outlet pipe shape on the flow fluctuation characteristics associated with a positive displacement hydraulic turbine. Renew Energy 2021;163:1046–62.
- [105] Hogan Timothy W, Cada Glenn F, Amaral Stephen V. The status of environmentally enhanced hydropower turbines. Fisheries 2014;39(4):164–72.
- [106] Furukawa Akinori, Shigemitsu Toru, Watanabe Satoshi. Performance test and flow measurement of contra-rotating axial flow pump. J Therm Stresses 2007;16(1):7–13.
- [107] Analysis of a 300 MW variable speed drive for pump-storage plant applications. In: 2005 European conference on power electronics and applications. 2005, p. 10.
- [108] Hunt Julian David, Zakeri Behnam, Lopes Rafael, Barbosa Paulo Sérgio Franco, Nascimento Andreas, Castro Nivalde José de, et al. Existing and new arrangements of pumped-hydro storage plants. Renew Sustain Energy Rev 2020;129:109914.
- [109] Javed Muhammad Shahzad, Ma Tao, Jurasz Jakub, Amin Muhammad Yasir. Solar and wind power generation systems with pumped hydro storage: Review and future perspectives. Renew Energy 2020;148:176–92.
- [110] Hoffstaedt JP, Truijen DPK, Fahlbeck J, Gans LHA, Qudaih M, Laguna AJ, et al. Low-head pumped hydro storage: A review of applicable technologies for design, grid integration, control and modelling. Renew Sustain Energy Rev 2022.
- [111] Seeanlagengesetz (SeeAnlG). 2016, URL https://www.gesetze-im-internet.de/ seeanlg/SeeAnlG.pdf.
- [112] Gesetz zur entwicklung und förderung der windenergie auf-see-gesetz WindSeeG). 2016, URL https://www.gesetze-im-internet.de/ windseeg/BJNR231000016.html.
- [113] Gesetz über die umweltverträglichkeitsprüfung (UVPG). 1990, URL https:// www.gesetze-im-internet.de/uvpg/UVPG.pdf.
- [114] The Council of European Communities. COUNCIL DIRECTIVE 92/43/EEC of 21 May 1992 on the conservation of natural habitats and of wild fauna and flora. 1992, URL https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX: 31992L0043&from=DE.
- [115] The Council of European Communities and European Parliament. Directive 2009/147/EC of the European parliament and of the council of 30 November 2009 on the conservation of wild birds -EUR-Lex - 32009L0147 - EN. 2009, URL https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=CELEX%3A32009L0147.
- [116] Gesetz über naturschutz und landschaftspflege (bundesnaturschutzgesetz -BNatSchG). 2009, URL https://www.gesetze-im-internet.de/bnatschg\_2009/ BNatSchG.pdf.

- [117] Gesetz zur ordnung des wasserhaushalts (wasserhaushaltsgesetz WHG). 2009, URL https://www.gesetze-im-internet.de/whg\_2009/WHG.pdf.
- [118] Marcelli Marco, Scanu Sergio, Frattarelli Francesco Manfredi, Mancini Emanuele, Carli Filippo Maria. A benthic zonation system as a fundamental tool for natural capital assessment in a marine environment: A case study in the northern tyrrhenian sea, Italy. Sustainability 2018;10(10):3786.
- [119] Schilt Carl R. Developing fish passage and protection at hydropower dams. Appl Animal Behav Sci 2007;104(3–4):295–325.
- [120] Beurskens J, De Haan S, Bauer P, Dieleman R. Dutch contribution to the discussion about a Taiwan integrated energy storage island (pre-feasibility exploration)". 2014.
- [121] European Parliament and the Council of the European Union. Directive 2014/52/EU of the European parliament and of the council of 16 april 2014 amending directive 2011/92/EU on the assessment of the effects of certain public and private projects on the environment text with EEA relevance. 2014.
- [122] IUCN. Biodiversity offsets. 2016, IUCN.
- [123] Piazzi L, Balestri E, Magri M, Cinelli F. Experimental transplanting of posidonia oceanica (L.) Delile into a disturbed habitat in the mediterranean sea. Bot Mar 1998;41(1–6):593–602.
- [124] Ng Chin Soon Lionel, Toh Tai Chong, Chou Loke Ming. Artificial reefs as a reef restoration strategy in sediment-affected environments: Insights from long-term monitoring. Aquat Conserv Mar Freshw Ecosyst 2017;27(5):976–85.
- [125] Van Duren LA, Gittenberger A, Smaal AC, Van Koningsveld M, Osinga R, van der Lelij JA Cado, et al. Rijke riffen in de Noordzee: verkenning naar het stimuleren van natuurlijke riffen en gebruik van kunstmatig hard substraat. Technical report, Deltares; 2016.
- [126] Lengkeek Wouter, Didderen K, Teunis M, Driessen F, Coolen JWP, Bos OG, et al. Eco-friendly design of scour protection: potential enhancement of ecological functioning in offshore wind farms: towards an implementation guide and experimental set-up. Bureau Waardenburg; 2017.
- [127] European Parliament and the Council of the European Union. Directive 2008/56/EC of the European parliament and of the council of 17 June 2008 establishing a framework for community action in the field of marine environmental policy (marine strategy framework directive) (text with EEA relevance). 2008.
- [128] Tidal lagoon power. Tidal lagoon.
- [129] Profish technology monitoring fish ecology.
- [130] Odeh Mufeed. A summary of environmentally friendly turbine design concepts. Technical report, U.S. Department of Energy Idaho Operations Office; 1999.
- [131] Natel Energy. Restoration hydro turbine. 2021, URL https://www.natelenergy. com/turbines/.
- [132] The "Salmon Come Back" campaign,
- [133] Association Saumon-Rhin Suivi des poissons migrateurs dans le Grand Est.
- [134] Mueller M, Pander J, Geist J. Evaluation of external fish injury caused by hydropower plants based on a novel field-based protocol. Fish Manag Ecol 2017;24(3):240–55.
- [135] Sadoul Bastien, Geffroy Benjamin. Measuring cortisol, the major stress hormone in fishes. J Fish Biol 2019;94(4):540–55.
- [136] Bell Alison M, Backström Tobias, Huntingford Felicity A, Pottinger Tom G, Winberg Svante. Variable neuroendocrine responses to ecologically-relevant challenges in sticklebacks. Physiol Behav 2007.
- [137] Lowry Christopher A, Moore Frank L. Regulation of behavioral responses by corticotropin-releasing factor. Gen Comp Endocrinol 2006;146(1):19–27.
- [138] Pankhurst NW. The endocrinology of stress in fish: An environmental perspective. Gen Comp Endocrinol 2011;170(2):265–75.
- [139] Ramakumar R, Butler Nicholas G, Rodriguez Alonso P, Venkata SS. Economic aspects of advanced energy technologies. Proc IEEE 1993;81(3):318–32.
- [140] Sheth Shabana, Shahidehpour Mohammad. Tidal energy in electric power systems. In: IEEE power engineering society general meeting, 2005. IEEE; 2005, p. 630–5.
- [141] Lee Chang-Hee, Lee Bum-Yeon, Chang Won Keun, Hong Seongjin, Song Sung Joon, Park Jinsoon, et al. Environmental and ecological effects of lake shihwa reclamation project in South Korea: a review. Ocean Coastal Manag 2014;102:545–58.
- [142] Kempener Ruud, Neumann Frank. Tidal energy technology brief. International Renewable Energy Agency (IRENA); 2014, p. 1–34.
- [143] Boretti Alberto. Trends in tidal power development. In: E3S web of conferences. Vol. 173. 2020, p. 01003.
- [144] Shen Wei, Chen Xi, Qiu Jing, Hayward Jennifier A, Sayeef Saad, Osman Peter, et al. A comprehensive review of variable renewable energy levelized cost of electricity. Renew Sustain Energy Rev 2020;133:110301.
- [145] Tran Thomas TD, Smith Amanda D. Incorporating performance-based global sensitivity and uncertainty analysis into LCOE calculations for emerging renewable energy technologies. Appl Energy 2018;216:157–71.
- [146] Baker AC. Tidal power. Thomas Telford Publishing; 1987.
- [147] Tidal energy ocean energy council". 2021.
- [148] Schmidt Oliver, Melchior Sylvain, Hawkes Adam, Staffell Iain. Projecting the future levelized cost of electricity storage technologies. Joule 2019;3(1):81–100.
- [149] Boer W, Consulting Kema, Verheij F, Zwemmer D, Bv Lievense, Das R, et al. The energy island–An inverse pump accumulation station. Vol. 6800. 2007.

# Renewable and Sustainable Energy Reviews 160 (2022) 112281

- [150] IRENA International Renewable Energy Agency. Renewable power generation costs in 2018. Technical report, 2019.
- [151] Alvarado-Ancieta CA. Kostenschätzung 2012 für die elektrische und mechanische ausrüstung des krafthauses in wasserkraft- und pumpspeicher-projekten. 2002, Wasserkraft & Energie.
- [152] SWECO Norge AS. Cost base for hydropower plants. 2012.
- [153] Connolly David. The integration of fluctuating renewable energy using energy storage. University of Limerick; 2011.
- [154] IRENA International Renewable Energy Agency. Renewable power generation costs in 2012: an overview. Technical report, 2013.
- [155] Oshima Katsuhiro, Kawai Juichiro, Otsuka Shiro, Wada Toshiro, Imano Haruo. Development of pump-turbine for seawater pumped storage power plant. In: Waterpower'99: hydro's future: technology, markets, and policy. 1999, p. 1–6.
- [156] Zakeri Behnam, Syri Sanna. Electrical energy storage systems: A comparative life cycle cost analysis. Renew Sustain Energy Rev 2015;42:569–96.